Author's response to reviews

Title: What are the perceived learning needs of Australian general practice registrars for quality prescribing?

Authors:

Rola Ajjawi (rola.ajjawi@monash.edu)
Jill E Thistlethwaite (j.e.thistlethwaite@warwick.ac.uk)
Parisa Aslani (parisa.aslani@sydney.edu.au)
Nick B Cooling (nick.cooling@gptt.com.au)

Version: 6 Date: 8 December 2010

Author's response to reviews:

Three changes were made:
1. the word [information] was added on page 11 to make more sense of the quote
2. brackets were replaced by commas on page 16
3. WHO changed to World Health Organization page 17